Author’s response to reviews

Title: The association of circulating irisin with metabolic risk factors in Chinese adults: a crosssectional community-based study

Authors:

Lizhi Tang (deepsky1219@126.com)
Yuzhen Tong (277930170@qq.com)
Fang Zhang (fun_906@hotmail.com)
Guilin Chen (1208572160@qq.com)
Yun Zhang (18280057628@163.com)
John Jobin (drjobinjohn@hotmail.com)
Nanwei Tong (tongnw@scu.edu.cn)

Version: 3 Date: 30 Nov 2019

Author’s response to reviews:

Dear editor,

Thank you very much your time and comments, we have updated our manuscript and made revisions according to your comments.

Assistant Editor Comments:

1. Overlap

-- We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in the Methods section of your manuscript.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.
Thank you for the comments, we have rephrased the expression to avoid the overlap.

2. **Availability of Data and Materials**

   -- The Availability of data and materials section refers to the raw data used in your study and therefore your current statement is not suitable.

   If you wish to share the data through correspondence, then please include the statement “The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.”

   We have updated it in the manuscript.

3. **Questionnaires**

   -- Please clarify whether the questionnaire used in your study was developed for this study or has previously been published elsewhere. If the questionnaire has been published elsewhere please provide a reference to it in your manuscript, if the questionnaire was developed for this study please upload an English language version as a supplementary file.

   We have added a reference which has previously been published in the manuscript.

4. **List of Abbreviations**

   -- Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

   We have added a list of abbreviation before the declaration section.

5. **Funding**

   -- In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

   The funding body had no role in the design of the study, or the collection, analysis, and interpretation of data, or in writing the manuscript.

6. **Authors’ Contributions**

   -- Currently, the contributions of authors [FZ, GLC, YCZ, JJ] do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

   An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from
Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship.

Anyone who contributed towards the article who does not meet the criteria for authorship can be acknowledged in the ‘Acknowledgements’ section.

We have provided further clarifications on their contributions for [FZ, GLC, YCZ, JJ]

7. Formatting

-- Please ensure your main manuscript conforms to the following headings: Background, Methods, Results, Discussion, Conclusions.

We ensure our main manuscript conforms to the headings mentioned above.

8. Clean Manuscript

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional SUPPLEMENTARY files are cited within the text.

We have uploaded the clean manuscript at this stage.